Author's response to reviews

Title: No effect of unemployment on Intimate Partner-Related Femicide during the financial crisis: a longitudinal ecological study in Spain

Authors:

Jordi Torrubiano-Domínguez (jordi.torrubiano@ua.es)  
Carmen Vives-Cases (carmen.vives@ua.es)  
Miguel San-Sebastián (miguel.sansebastian@epiph.umu.se)  
Belén Sanz.Barbero (bsanz@iscii.es)  
Isabel Goicolea (isabel.goicolea@epiph.umu.se)  
Carlos Álvarez-Dardet (carlos.alvarez@ua.es)

Version: 4  
Date: 19 September 2015

Author's response to reviews:

Dear editor,

we have changed the Authors' Contribution section in order to conform the manuscript to the journal style.

You can see this changes remarked in red in lines 293-97.

I also want to thank you for the revision process.

Best Regards,

Jordi Torrubiano

University of Alicante